

Citation: Brückner S, Agnandji ST, Berberich S, Bache E, Fernandes JF, Schweiger B, et al. (2015) Effect of Antihelminthic Treatment on Vaccine Immunogenicity to a Seasonal Influenza Vaccine in Primary School Children in Gabon: A Randomized Placebo-Controlled Trial. PLoS Negl Trop Dis 9(6): e0003768. doi:10.1371/journal.pntd.0003768

Editor: Gary L. Simon, George Washington University, UNITED STATES

Received: December 8, 2014

Accepted: April 17, 2015

Published: June 8, 2015

Copyright: © 2015 Brückner et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information file.

Funding: The study was funded by the Bundesministerium für Bildung und Forschung (BMBF). The grant number is 01 KA 1009. The ministery can be found under the following url <u>http://</u> <u>www.bmbf.de/</u>. The author who received funding was ME. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. **RESEARCH ARTICLE**

Effect of Antihelminthic Treatment on Vaccine Immunogenicity to a Seasonal Influenza Vaccine in Primary School Children in Gabon: A Randomized Placebo-Controlled Trial

Sina Brückner^{1,2}, Selidji T. Agnandji^{1,2}, Stefan Berberich², Emmanuel Bache², José F. Fernandes², Brunhilde Schweiger³, Marguerite Massinga Loembe², Thomas Engleitner^{1^a}, Bertrand Lell^{1,2}, Benjamin Mordmüller^{1,2}, Ayola A. Adegnika^{1,2,4}, Maria Yazdanbakhsh⁴, Peter G. Kremsner^{1,2}, Meral Esen^{1,2}*

1 Institute of Tropical Medicine, University of Tübingen, Tübingen, Germany, 2 Centre de Recherches Médicales de Lambaréné, Albert Schweitzer Hospital, Lambaréné, Gabon, 3 Nationales Referenzzentrum für Influenza, Robert-Koch-Institut, Berlin, Germany, 4 Leiden Medical University Center, Department of Parasitology, Leiden, The Netherlands

¤ Current address: Department of Medicine II, Klinikum Rechts der Isar, Technische Universität München, München, Germany

* meral.esen@uni-tuebingen.de

Abstract

Background

Helminth infections are a major public health problem, especially in the tropics. Infected individuals have an altered immune response with evidence that antibody response to vaccination is impaired. Hence, treatment of helminth infections before vaccination may be a simple intervention to improve vaccine immunogenicity. In the present study we investigated whether a single-dose antihelminthic treatment influences antibody responses to a seasonal influenza vaccine in primary school children living in Gabon, Central Africa.

Methods

In this placebo-controlled double-blind trial conducted in Gabon the effect of a single-dose antihelminthic treatment with 400 mg albendazole versus a placebo one month prior to immunization with a seasonal influenza vaccine was investigated. Antiviral antibody titers against all three vaccine strains were assessed by haemagglutination inhibition (HI) test at baseline (Day 0; vaccination) and four weeks (Day 28) as well as 12 weeks (Day 84) following vaccination. Vaccine-specific memory B-cell response was measured at Day 0 and Day 84 by vaccine-specific Enzyme-linked Immunospot (ELISpot) assay. The trial is registered with the Pan African Clinical Trials Registry (PACTR) (PACTR201303000434188).

Competing Interests: The authors have declared that no competing interests exist.

Results

98 school children aged 6–10 years were randomly allocated to receive either antihelminthic treatment or placebo and were vaccinated one month after the treatment. The prevalence of helminths at baseline was 21%. Vaccine-specific HI titers against at least one of the three vaccine strains increased at Day 28 and Day 84 in all participants. HI titers against both influenza A strains as well as memory B-cell response were modestly higher in the antihelminthic treated group compared to the placebo group but the difference was not statistically significant. Total but not specific IgA was elevated in the antihelminthic treated group compared to the control group at Day 28.

Conclusion

In our setting antihelminthic treatment had no significant effect on influenza vaccine immunogenicity. A trend towards better antiviral and vaccine immunogenicity in the antihelminthic treated group encourages studies to be conducted with alternative treatment schedules or in populations with a higher helminth burden.

Author Summary

Helminth infections are a major health problem in the tropics and most affected are children. The parasites are able to influence the immune system from a T-helper 1 type response to a T-helper 2 type response. There is evidence that in infected individuals the immune response following vaccination is impaired. Thus pre-treatment with a singledose of an antihelminthic treatment before vaccination could be a simple and cost-effective intervention to improve vaccine efficacy. In the present study we investigated whether a single-dose antihelminthic treatment with albendazole influences the vaccine outcome to a seasonal influenza vaccine in primary school children living in Gabon, Central Africa. We observed a trend towards a higher anti-viral antibody titer after vaccination in the pretreated group compared to the placebo control group, albeit not statistical significant. Furthermore we detected a higher concentration of total IgA but not of vaccine-specific IgA. In conclusion, our findings show subtle effects of antihelminthic pre-treatment but are not conclusive enough to recommend a single-dose of albendazole before vaccination to improve vaccine immunogenicity but encourage to conduct further studies in endemic areas with other treatment regiments.

Introduction

Infection with geohelminths, mainly *Ascaris (A.) lumbricoides, Trichuris (T.) trichiura* and hookworm, is a major public health problem affecting 20% of the world's population, mainly for those living in Sub-Saharan Africa (SSA). As access to public health programs is widely lacking, geohelminthiasis is considered by the World Health Organization (WHO) as one of the most neglected tropical diseases with serious health, nutritional and social outcomes for the affected individuals [1–3]. Vulnerable groups are children [2] and pregnant women [3]. Chronic infection with geohelminths has an impact on health as well as on cognitive skills [4–8] and it is known that infection with helminths leads to immune response alterations. Usually, T-helper type 2 (Th2) immune responses [9–12] are predominant and a general suppression of innate

and adaptive T- and B-cell responses via the activation of regulatory T-cells (Treg) and/or induction of anti-inflammatory cytokines[<u>10,13,14</u>] may lead to general hyporesponsiveness of the immune system[<u>14,15</u>].

Vaccination is one of the most effective tools to prevent infectious diseases. Nonetheless seroconversion and therefore efficacy are variable in vaccinated individuals depending on age, environment and genetic host factors[16–18]. In addition, acute and chronic infections have an influence on vaccine outcome[19,20]. The interaction of geohelminth infection and vaccination is not well investigated although immunization programs for infants are well implemented in areas where geohelminths are highly endemic. Until now it has been shown that *A. lumbricoides* has an impact on the immune response induced by an oral cholera vaccine[21] and that intestinal parasites influence the outcome of a Bacillus-Calmette-Guérin (BCG) vaccination [22]. There is evidence that the presence of geohelminths, especially *T. trichiura* negatively influences immune responses against GMZ2, a malaria vaccine candidate[23].

Past studies from Gabon showed that Gabonese school children are heavily infected with intestinal parasites (infection rates of *A. lumbricoides* 46%, or *T. trichiura* 71%) and 74% of examined children were at least positive for one of the investigated helminths[24–26]. Regular antihelminthic treatment in high-risk groups like school children is considered as an effective tool for controlling the burden of geohelminth infection but is not widely implemented in Gabon. The WHO promotes helminth control by periodic deworming once or twice a year, depending on prevalence, as a cost-effective intervention[27,28]. However regular deworming is not yet implemented in all endemic countries[29].

Antihelminthic treatment would be a cost-effective and easy tool to reduce worm burden [<u>30</u>] and may simultaneously increase vaccine immunogenicity.

Therefore in the present study we investigated the effect on vaccine immunogenicity of pretreatment with a single-dose of albendazole four weeks prior to a scheduled seasonal influenza vaccination in Gabonese primary school children.

Materials and Methods

Trial design and setting

For this double-blinded randomized trial healthy primary school children from Lambaréné and surroundings were randomized to receive either antihelminthic treatment (albendazole 400 mg) (Micro Lab ltd, India) or placebo (Laboratories Sterop, Belgium) four weeks (Day -28) prior to vaccination with either a seasonal influenza vaccination (VAXIGRIP, Sanofi Pasteur, season 2011/2012) intra muscularly (i.m.) (n = 98), Polysaccharide Meningococcal A+C vaccine (Sanofi Pasteur) sub cutaneously (s.c.) (n = 104) or an oral cholera vaccine (Dukoral, Sanofi Pasteur) (n = 106) administered at Day 0. Vaccinations were given in three subsequent time slots. The first cohort of primary school children was vaccinated with the influenza vaccine, the second with meningococcal vaccine and the third cohort received two times the cholera vaccine.

Inclusion criteria were age from 6 to 10 years (primary school children), a signed informed consent form (ICF), good general health upon clinical examination and no acute symptoms of geohelminths infection. Furthermore the participants and their legal representative was asked if he/she will be resident in the area until the end of the study.

Exclusion criteria were the participation in another clinical trial, known contraindication to antihelminthic treatment or to the administration of one of the chosen vaccines including i.m. or s.c. administration, known immunization against the vaccine antigens, known infection with pathogens of one of the vaccine antigens in the past except for influenza (because the influenza vaccine strain composition is different each year), any confirmed or suspected

immunosuppressive or immunodeficient condition resulting from disease (e.g., malignancy, HIV infection) or immunosuppressive/cytotoxic therapy as well as acute disease at the beginning of the study and before vaccination. If a child was febrile at the scheduled time of vaccination, injections were postponed until the infection was cured. If a child had a known acute or chronic disease like malaria, AIDS or tuberculosis as well as a haemoglobin level < 7 g/dl or signs of haematuria and/or proteinuria tested by urine sticks (Combur 9 test) the child was not included and referred to the Albert Schweitzer Hospital (ASH) for treatment. If menarche was reported a pregnancy test was be performed at Day -28 and Day 0 prior to antihelmintic treatment and prior to vaccination.

Here, we report results of the first part of the study, where the children were vaccinated with the seasonal influenza vaccine.

Children, who were infected with *Schistosoma* (*S*.) *haematobium*, as well as children with any other symptomatic infection were excluded from the study and treated accordingly. All parasite positive participants (including those without symptoms) received appropriate treatment after study termination.

Immunological investigations

Blood was taken on the day of vaccine injection (Day 0), Day 28 (four weeks after vaccination) and Day 84 (12 weeks after vaccination). The primary immunological endpoint of the study was functional antibody level measured as haemagglutination inhibition (HI) testing. To assess memory B-cell response (secondary immunological endpoint) against the vaccine antigens, an Enzyme-linked Immunospot (ELISpot) assay was performed.

Assessment of anti-influenza antibody titers

Pre- and post-vaccination samples were analyzed by a validated microtiter haemagglutination inhibition (HI) test at the German National Reference Center for Influenza (NRZ Influenza, Robert Koch Institute (RKI), Berlin) as previously described[<u>31</u>]. Prior to testing, each serum was treated with receptor-degrading enzyme to inactivate non-specific inhibitors at a final serum dilution of 1:10. Sera were then diluted serially two fold into microtiter plates. Each virus strain was adjusted to 4 HA units/25 μ l which was verified by back titration and 25 μ l of this virus suspension was added to each of the 96 wells. After incubation at room temperature (RT) for 30 min freshly prepared 0.5% turkey red blood cells (RBCs) were added, the plates were mixed, followed by a further incubation at RT for 30 min. Human sera serving as positive controls and negative controls were included on each plate. HI titers were reported as the reciprocal of the last serum dilution that contained non-agglutinated RBCs.

Assessment of total immunoglobuline (Ig) isotypes and subclasses

Pre- and post-vaccination samples (Day 0, Day 28 and Day 84) were analyzed by using a Multiplex assay (Biorad, Germany) for detection of multiple antibodies, like IgG subclasses (IgG1; IgG2, IgG3 and IgG4), total IgE, total IgM and total IgA at baseline, Day 28 and Day 84 post vaccination. The assay was conducted according to the manufacturer's specifications.

Assessment of vaccine-specific IgA by enzyme-linked immunosorbant assay (ELISA)

To assess vaccine-specific IgA concentrations plates (Nunc, Germany) were coated with 0.5 μ g/ml of the vaccine antigen, incubated over night at 4°C and blocked with blocking buffer (0.3% milkpowder (Roth, Germany), 0.1% Tween-20 (Sigma, Germany) and PBS (Life

technologies, USA) for 1 h at RT. Samples were plated in serial dilutions for 2 h at RT. As secondary antibody a polyclonal rabbit anti-human IgA/HRP (Dako, Germany) was used. For visualization TMB one (Kem En Tec, Denmark) and H_2SO_4 (Merck, Germany) was used. OD was measured using a Photometer (Phomo, Anthos, Germany) at wave length 450 nm and 620 nm as reference.

Vaccine-specific memory B-cell ELISpot

As previously described, peripheral blood mononuclear cells (PBMCs) were frozen on Day 0 and Day 84 and later used for memory B-cell ELISpot[32,33]. In brief, PBMC were separated from heparinized full blood by gradient centrifugation (Ficoll-Plaque PLUS, GE Healthcare, Sweden), counted and frozen in 90% fetal calf serum (FCS Gold PAA, Germany) and 10% DMSO (Sigma, Germany) at -150°C. Before ELISpot, cells were thawed, counted and seeded at a density of 1*10⁶ cells per ml in RPMI 1640 (Sigma, Germany), complemented with sodium pyruvate, non-essential amino acids, L-glutamine, penicillin, streptomycin (all supplemented from Life Technologies, USA) and 10% heat inactivated FCS (FCS Gold, PAA, Germany). Maturation of circulating memory B-cells into antibody-secreting cells (ASC) was performed by in vitro stimulation with 2.5 µg/ml CpG-2006 (TIB-MOLBIOL, Germany) and 10 ng/ml recombinant human IL-15 (R&D systems, USA) in 24 wells cell culture plates (Corning Costar) for 6 days at 37°C, 5% CO₂ as described [34]. Following maturation, 2^*10^5 cells were serially diluted on Ag-specific 96 well plates (Millipore, Germany). The vaccine (VAXIGRIP) was coated over night at a concentration of 5µg/ml in PBS. Plates were washed and blocked for 1 hour (h) with complete medium. Meanwhile cells were washed and counted. After seeding cells were incubated at 37°C, 5% CO₂ for 2 h allowing them to secrete antibodies. Secreted antigen (Ag)-specific antibodies were detected using biotin labeled anti-human IgG antibody (Sigma, Germany, 1:500 in PBS, 3% BSA) and ExtrAvidin peroxidase (SIGMA, Germany, 1:600 in PBS, 5% BSA). AEC substrate (SIGMA, Germany) was added for 10 min at RT. After staining the plates were dried over night. Spots were counted by the CTL ImmunoSpot (CTL, USA).

Assessment of parasite burden

Stool samples were collected at Day -28, Day 0 and Day 84 and analyzed by using the qualitative Merthiolate-Iodine-Formaldehyde (MIF)-technique[35]. In brief, 10 ml of Merthiolate-Formaldehyde-Solution (5% Formaldehyd, 1% Glycerine (Merck, Germany)) and Lugol's Solution (10% potassium iodine, 5% iodine (Merck, Germany)) were added to each walnut size stool sample (approx. 5 g) filtered through a metal wire, centrifuged for 5 min 1500 rounds per minute (rpm) and analyzed by microscopy at the end of the study.

Urine was examined for the presence of *S. haematobium* by urine filtration method[<u>36</u>] using 10 ml of well mixed urine passed through a filter (12 μ m pore size, Millipore, Germany). The filter was transferred to a glass slide, stained with methylene blue and analyzed by microscopy.

At Day 0 and Day 84 thick blood smears were performed to assess malaria parasites retrospectively. If a child presented with fever or any other symptom suggestive of malaria, a rapid test (Paracheck Pf) was performed and the volunteer was treated with appropriate treatment.

Sample size calculation

We assumed that a 30% difference of immune responses between antihelminthic treatment and placebo groups is clinically relevant. In a pilot trial with the malaria vaccine candidate GMZ2 we observed a difference in antibody response greater than 30% with a standard deviation of 0.5[23]. A sample size of 45 individuals (n = 45) per group is required to detect such a difference with a power of 80% at a significance level of 0.05. To allow for 15% of loss to follow-up a total of 52 (n = 52) schoolchildren per group were required. Sample size calculation was done using R v2.9.0[<u>37</u>].

Randomization, data entry and statistical analysis

The randomization and data management was performed on the "Koordobas" database system (Institute for Clinical Epidemiology and applied Biometry, University of Tübingen) The medical personal treating participants were not involved in the outcome evaluation. Group differences at follow-up visits (Day 28, Day 84) were determined with a rank based ANCOVA (Analysis of Covariance) corrected for baseline titer and gender. Differences in ELISpot counts were assessed by a Wilcoxon test. Geometric mean titers were calculated according to $10^{mean(log_{10}(1+HI-titer))}$. A two sided alpha of 0.05 was used as significance level.

Ethical approval

The study was approved by the Comité d'Ethique Régional Indépendant de Lambaréné and the Comité National d'Ethique pour la Recherche du Gabon. The trial is registered with PACTR (PACTR201303000434188) and the study was conducted in accordance to the Declaration of Helsinki and followed the International Conference on Harmonization (ICH) Good Clinical Practice (GCP) guidelines. The child's parents or their legally accepted representatives provided written informed consent before study participation.

Results

From December 2011 until September 2012 out of 113 screened primary school children from Lambaréné and surroundings 98 were randomized to receive either antihelminthic treatment (n = 50) or placebo (n = 48). 92 participants were vaccinated with a seasonal influenza vaccine and 82 terminated the study with visit 4 at Day 84 post vaccination (Fig_1). Baseline characteristics were similar between the two groups except that the fraction of females was higher in the antihelminthic treated group (57% versus 34% in the placebo group) (Table 1). A possible effect of gender was taken into account in exploratory and sensitivity analyses.

Anti-influenza antibody titers

Antibodies against the three influenza vaccine strains A/California/7/09 (A(H1N1)pdm09), A/ Perth/16/09 (A(H3N2)) and B/Brisbane/60/08 were determined by HI testing. HI titers increased against at least one vaccine strain after vaccination at Day 28 and Day 84 in all participants (<u>Fig 2</u>). The increase of HI values for A(H1N1)pdm09 and A(H3N2) tended to be higher in the antihelminthic treated group compared to the control group but the difference was not statistically significant (Figs <u>2</u> and <u>3</u>).

Thirty-four participants already had detectable antibodies at baseline for the A(H1N1) pdm09 strain. The baseline titers ranged between 15 and 640 with a median of 120 in the antihelminthic treated group and 80 in the placebo group and the GMT was 28 in the antihelminthic treated group and 15 in the control group; 19 participants in the antihelminthic treated group and 17 participants in the placebo group were without a detectable baseline HI titer. At Day 28 the HI titers increased up to 5000 (median: 640 in both groups and a GMT of 134 in the antihelminthic treated group and 84 in the control group,) and decreased until Day 84 (median: 320 in both groups and GMT of 138 in the antihelminthic treated group and 94 in the control group) (Tables <u>2</u> and <u>3</u>).



Fig 1. Study profile. *Patients were excluded, because of infection with *S. haematobium.* *Participants not terminating the study are summarized as lost to follow up (n = 16).

Antibody titers against the strain A(H3N2) ranged from 15 to 160 (median of 120 in the antihelminthic treated group and 80 in the control group), whereas the GMT was 73 for the antihelminthic treated and 72 for the placebo group at Day 0. For this strain also an increase of antibodies at Day 28 with a median of 640 in both groups and a GMT of 554 in the antihelminthic treated group and 516 in the placebo group as well as decreasing values at Day 84 with an median of 320 in both groups and a GMT of 344 in the antihelminthic treated group and 258 in the control group were observed (Tables 2 and 3).

Antibodies against the influenza B vaccine strain had a median of 320 at Day 28 in both groups and declined until Day 84 with a median of 160 in both groups. GMT at Day 28 was 142 in the antihelminthic treated group and 164 in the placebo group and at Day 84 120 in the antihelminthic treated group and 97 in the control group. At baseline the median was 0 in both groups and the GMT was 12 for the antihelminthic treated group and 10 for the placebo group (Tables <u>2</u> and <u>3</u>).

Assessment of total Ig isotypes and subclasses

Ig isotypes and subclasses (IgG1-4, IgA, IgE and IgM) pre- and post-vaccination were assessed by a multiplex system. Total IgA was elevated in the antihelminthic treated group compared to the control group at Day 28 (p-value 0.006) (Fig 4).

	Antihelminthic treatment	Placebo
Gender		
Male	22 (22.4%)	32 (32.7%)
Female	28 (29.6%)	16 (16.3%)
Age		
6 years	15 (15.3%)	16 (16.3%)
7 years	9 (9.2%)	8 (8.2%)
8 years	14 (14.3%)	12 (12.2%)
9 years	10 (10.2%)	12 (12.2%)
10 years	2 (2%)	0
Mean age	7.43	
Helminth		
Single infection		
A. lumbricoides	2 (2%)	2 (2%)
F. hepatica	0	2 (2%)
T. trichiura	4 (4%)	3 (3%)
Multiple infection		
T. trichiura/ A. lumbricoides	2 (2%)	4 (4%)
T. trichiura/ A. lumbricoides/ A. duodenale	0	1 (1%)
Negative	41 (41.84%)	33 (33.3%)
NA	1 (1%)	3 (3%)

Table 1.	Baseline	characteristics	and helminth	infection at da	v -28.
----------	----------	-----------------	--------------	-----------------	--------

The total concentration of the subclasses IgG1 and IgG3 were slightly elevated in the antihelminthic treated group (for IgG1 p-value was 0.042 and for IgG3 p-value was 0.03 in the model-based analysis, but was not significant using the Wilcoxon-test (p-value 0.347 and 0.160)).

To evaluate whether the elevation of total IgA indicates a higher amount of vaccine-specific IgA we performed a vaccine-specific ELISA. Here, no difference of vaccine-specific IgA was detected between the antihelminthic treated and control group ($\underline{Fig 5}$).

Assessment of ASC

B-cell ELISpot assay to measure vaccine-specific memory B-cells was performed with samples from 75 individuals.

Antigen-specific memory B-cells were detectable in all vaccinated subjects at Day 84 (Fig_6). Median number of vaccine-specific ASC per 100,000 PBMC was 13 at Day 84 (range: 1 to 144) in all participants.

In 10 participants of the antihelminthic treated group and 12 participants of the placebo group a low number of ASCs (range: 0 to 47) was already detected at Day 0 (Fig 6). The number of detectable ASCs in the antihelminthic treated group at Day 84 ranged from 0 to 240 (median 19) and from 0 to 140 (median 3) in the placebo group (Fig 7). Nonetheless this difference was not statistically significant.

Assessment of parasite burden

The percentage of helminth burden in our setting was 21%. From these 21%, infection with *A*. *lumbricoides* and *T. trichiura* was 6% for each species. The burden did not differ between the visits (Tables 1 and 4).



Fig 2. Antibody titers against the three vaccine strains at baseline (day 0), day 28 and day 84. Red lines indicate the mean of all volunteers of the antihelminthic treated group (AT) and blue lines indicate the mean of all participants of the placebo group. Dashed lines indicate antibody titers of each participant.



California (H1N1) Perth (H3N2) Brisbane (Influenza B)

Fig 3. Differences of HI titers between the respective visits (day 28, day 84) and day 0 (baseline). Red and blue colors represent the pre-treated (AT) and control group.

doi:10.1371/journal.pntd.0003768.g003

Table 2. Median, 25% and 75% quartile of vaccine strain specific HI titers.

	Day 0	Day 28	Day 84
A(H1N1)pdm09			
Antihelminthic treatment	120 (0,160)	640 (0,1280)	320 (35,960)
Placebo	80 (0,60)	640 (0,960)	320 (20,480)
A(H3N2)			
Antihelminthic treatment	120 (20,240)	640 (320,960)	320 (280,640)
Placebo	80 (40,160)	640 (480,1280)	320 (240,600)
Influenza B			
Antihelminthic treatment	0 (0,40)	320 (120,320)	160 (80,320)
Placebo	0 (0,20)	320 (120,520)	160 (80,800)

doi:10.1371/journal.pntd.0003768.t002

At Day 0 and Day 84 9 and 10 volunteers were positive in the thick blood smear, respectively.

Discussion

Recent studies suggest that infection with helminths influences the immunological outcome of vaccination. A study recently conducted in rural Gabon showed that children infected with helminths had an impaired antibody response against an influenza vaccine compared to those free of infection[14]. During a phase Ib trial investigating immunogenicity of the malaria vaccine candidate GMZ2 in Gabonese children those infected with *T. trichiura* exhibited a lower antibody response against the vaccine antigens compared to those who were not infected with the parasite[23]. Since these and other studies in animal models and humans [21,38,39] show that helminths and other intestinal infections negatively influence vaccine immunogenicity we conducted a study to test the hypothesis that antihelminthic treatment prior to vaccination will increase immune responses to vaccine antigens.

Because antibody response towards vaccine antigens is a surrogate marker for protection in areas where infectious diseases are highly endemic an effective immune response to vaccinated antigens is very important $[\underline{40}-\underline{42}]$.

In the present study anti-viral antibody response was analyzed by HI test for each of the three influenza strains administered with a seasonal vaccine. The influenza vaccine was selected because a single-dose is sufficient to raise antibody responses towards a protective titer and the vaccine is not part of the Expanded Program on Immunization (EPI) in Gabon. Therefore, no or only low baseline titers would presumably be present in the study population. As expected

Table 3. GMT of vaccine strain specific antibodies.

	Day 0	Day 28	Day 84
A(H1N1)pdm09			
Antihelminthic treatment	28	134	138
Placebo	15	84	94
A(H3N2)			
Antihelminthic treatment	73	554	344
Placebo	72	516	258
Influenza B			
Antihelminthic treatment	12	142	120
Placebo	10	164	97

doi:10.1371/journal.pntd.0003768.t003





there was an increase of HI titers against each vaccine strain after vaccination at Day 28 and Day 84 in all participants. This was observed mainly for the antibody concentration against the influenza A strains A(H1N1)pdm09 and A(H3N2). The low but frequently seen baseline HI titers against the influenza A (H1N1) and A(H3N2) in some participants suggest that these children were already in contact with circulating virus strains or that they have cross-reactive antibodies from recent circulating influenza strains or from other cross-reactive pathogens. In the present study we assessed that total IgA concentrations were higher in individuals of the antihelminthic treated group compared to the control group four weeks following vaccination. Because IgA is crucial for the control of influenza[43-45] and vaccine-specific IgA can be detected in mice following influenza vaccination [46] we also measured vaccine-specific IgA but saw no difference between the groups. Since total IgA was not different at baseline we do not know if the difference of total IgA at Day 28 is an effect of vaccination or a late effect of the antihelminthic treatment. This effect should be further investigated in more detail in particular because IgA has an important function during the defense of airborne and gastrointestinal infections. However in our setting we have not examined the role of this finding and we have not determined secretory IgA.

Besides we could show that the number of vaccine-specific ASCs representing memory Bcells was elevated in the antihelminthic treated group compared to the control group but this difference did also not reach statistical significance. We assume that the already existing ASCs at baseline in samples of some participants are due to previous influenza infections. Since the antibody titers as well as the number of ASCs were not significantly elevated in the antihelminthic treated group the question arises if a better or more effective antihelminthic treatment







would have had clearer effects on the vaccine immunogenicity. The fact that the overall helminth burden did not differ between the visits implies that single-dose antihelminthic treatment is not sufficient to cure or to prevent relevant helminth infection or that reinfection occurs rapidly. This becomes more evident since Adegnika et al. very recently showed that short antihelminthic treatment regimens are efficacious to cure A. lumbricoides but not T. trichiura infection. To eliminate T. trichiura infection at least a double treatment seems to be necessary[30]. Therefore in our setting the treatment regimen was not adequate to eliminate helminths sufficiently and to reconstitute the immune system. Cooper et al. investigated the effect of a double-dose (2x200mg) albendazole treatment prior to an oral cholera vaccination leading to a highly significant increase in anti-vibriocidal antibodies[21,47]. In contrast to the present study, the investigators only included individuals carrying A. lumbricoides[21,30]. In our setting we retrospectively analyzed the helminth burden of the volunteers at the end of the study and noticed that the number of infections amongst primary school children was not as high as originally suggested from recent studies, which reduces the power of our analyses. In a study performed 2004 the overall prevalence of helminth infection was 74%[24], whereas in our study population the worm burden was only 21%. This could be due to different reasons. First, in the study conducted by van den Biggelaar et al. the Kato Katz method to detect the egg load in fresh stool samples was used [24] whereas we used the MIF technique to detect the worm burden in preserved stool samples at the end of the study to ensure all investigators were blind to the infection status of the participants. The Kato Katz method has a higher sensitivity [48] compared to the MIF-technique [49] and this may explain partially the unexpected low prevalence of helminth infection although other studies using the same methodology gave



higher prevalence rates. Besides the possibility that it is a chance finding the reduction of worm burden could be due to better public health facilities and better education of the population as well as to mass drug administration (MDA) even when it is administered on an irregular basis or by uncontrolled self-mediation. Gabon is a highly endemic country for STH according to the WHO, therefore MDA is recommended but in the region where the study was performed it is not regularly administered[29]. Private use is difficult to assess since individuals do not need prescription to buy antihelminthic treatment over-the-counter[50,51] and often the individuals do not report the use of antihelminthic treatment. Furthermore it could be that a member of a household was recently treated and therefore also infection of other family members decreased [52]. Taken all this into account our in our study population the helminth burden was lower as expected for reasons which were not elicited in this study and our sample size was not powered for such a low helminth prevalence.

In conclusion, we showed that in our setting there was non-significant difference in virusspecific HI titers and ASCs against the vaccine antigens between the antihelminthic treated and the placebo group. Furthermore at Day 28 post vaccination total IgA was higher in the antihelminthic treated group compared to the control group. But there was no difference comparing vaccine-specific IgA titers. We can only speculate if a single dose antihelminthic treatment is sufficient to increase vaccine immunogenicity in a setting with higher helminth prevalence or if an appropriate more powerful treatment could contribute to a better immune response to vaccination. This has to be investigated in more detail and with different antihelminthic regiments.





Day 0

Day 84

Fig 7. Vaccine-specific IgG ASCs at day 0 and day 84, in antihelminthic treated (AT) and placebo group. Red and blue represent antihelminthic treated (AT) and control group.

doi:10.1371/journal.pntd.0003768.g007

Table 4. Distribution of the worm burden in the two groups at day 0 and day 84.

	Day 0		Day 84	
	Antihelminthic treatment	Placebo	Antihelminthic treatment	Placebo
Single infection				
A. lumbricoides	2 (2.2%)	2 (2.2%)	5 (6.1%)	4 (4.9%)
T. trichiura	2 (2.2%)	3 (3.3%)	3 (3.6%)	1 (1.2%)
S. haematobium	0	0	2 (2.4%)	1 (1.2%)
Taenia	0	0	0	1 (1.2%)
Tapeworm	1 (1.1)	0	0	0
Multiple infection				
T. trichiura/ A. lumbricoides	1 (1.1%)	2 (2.2%)	0	3 (3.7%)
S. haematobium/ T. trichiura	0	0	0	1 (1.2%)
S. haematobium/ A. lumbricoides	0	0	1 (1.2%)	0
T. trichiura/ A. duodenale	1 (1.1%)	0	0	0
T. trichiura/A. lumbricoides/ A. deodenale	0	1 (1.1%)	1 (1.2%)	0
Neg	39 (42.4%)	32 (34.7%)	25 (30.5%)	24 (29.3%)
NA	2 (2.2%)	4 (4.4%)	7 (8.5%)	3 (3.7%)

doi:10.1371/journal.pntd.0003768.t004

Supporting Information

S1 Checklist. Consort checklist. (PDF)

Acknowledgments

The authors thank the NRZ Influenza team of the RKI for performing the HI assays, Sonja Killinger for logistics and data entry, the staff of CERMEL and all participants as well as the members of the Institute for Clinical Epidemiology and applied Biometry in Tübingen.

Author Contributions

Conceived and designed the experiments: ME STA PGK MY. Performed the experiments: SBr ME STA SBe EB JFF BS TE BM MML. Analyzed the data: SBr ME TE BM. Contributed reagents/materials/analysis tools: ME STA BL BM AAA BS. Wrote the paper: SBr ME BM STA BS TE BL AAA MY PGK.

References

- 1. Hall A. Micronutrient supplements for children after deworming. Lancet Infect Dis. 2007; 7: 297–302. PMID: <u>17376387</u>
- Ezeamama AE, Friedman JF, Acosta LP, Bellinger DC, Langdon GC, Manalo DL, et al. Helminth infection and cognitive impairment among Filipino children. Am J Trop Med Hyg. 2005; 72: 540–548. PMID: 15891127
- Matthys B, Bobieva M, Karimova G, Mengliboeva Z, Jean-Richard V, Hoimnazarova M, et al. Prevalence and risk factors of helminths and intestinal protozoa infections among children from primary schools in western Tajikistan. Parasit Vectors. 2011; 4: 195. doi: <u>10.1186/1756-3305-4-195</u> PMID: <u>21981979</u>
- Mascarini-Serra L. Prevention of soil-transmitted helminth infection. J Glob Infect Dis. 2011; 3: 175– 182. doi: <u>10.4103/0974-777X.81696</u> PMID: <u>21731306</u>
- Taylor-Robinson DC, Maayan N, Soares-Weiser K, Donegan S, Garner P. Deworming drugs for soiltransmitted intestinal worms in children: effects on nutritional indicators, haemoglobin and school performance. Cochrane Database Syst Rev. 2012;
- Cooper PJ, Chico ME, Guadalupe I, Sandoval CA, Mitre E, Platts-Mills TAE, et al. Impact of early life exposures to geohelminth infections on the development of vaccine immunity, allergic sensitization, and allergic inflammatory diseases in children living in tropical Ecuador: the ECUAVIDA birth cohort study. BMC Infect Dis. 2011; 11: 184. doi: <u>10.1186/1471-2334-11-184</u> PMID: <u>21714922</u>
- Jardim-Botelho A, Raff S, DeÁvila Rodrigues R, Hoffman HJ, Diemert DJ, Corrêa-Oliveira R, et al. Hookworm, Ascaris lumbricoides infection and polyparasitism associated with poor cognitive performance in Brazilian schoolchildren. Trop Med Int Health. 2008; 13: 994–1004. doi: <u>10.1111/j.1365-3156.</u> <u>2008.02103.x</u> PMID: <u>18627581</u>
- Lobato L, Miranda A, Faria IM, Bethony JM, Gazzinelli MF. Development of cognitive abilities of children infected with helminths through health education. Rev Soc Bras Med Trop. 2012; 45: 514–519. PMID: <u>22930049</u>
- 9. Maizels RM, Balic A, Gomez-Escobar N, Nair M, Taylor MD, Allen JE. Helminth parasites—masters of regulation. Immunol Rev. 2004; 201: 89–116. PMID: <u>15361235</u>
- Maizels RM, Yazdanbakhsh M. Immune Regulation by helminth parasites: cellular and molecular mechanisms. Nat Rev Immunol. 2003; 3: 733–744. PMID: <u>12949497</u>
- Dauby N, Goetghebuer T, Kollmann TR, Levy J, Marchant A. Uninfected but not unaffected: chronic maternal infections during pregnancy, fetal immunity, and susceptibility to postnatal infections. Lancet Infect Dis. 2012; 12: 330–340. doi: <u>10.1016/S1473-3099(11)70341-3</u> PMID: <u>22364680</u>
- 12. Moreau E, Chauvin A. Immunity against Helminths: Interactions with the Host and the Intercurrent Infections. J Biomed Biotechnol. 2010; 2010: 1–10.
- **13.** Bourke CD, Maizels RM, Mutapi F. Acquired immune heterogeneity and its sources in human helminth infection. Parasitology. 2010; 138: 139–159. doi: <u>10.1017/S0031182010001216</u> PMID: <u>20946693</u>

- 14. Van Riet E, Hartgers FC, Yazdanbakhsh M. Chronic helminth infections induce immunomodulation: consequences and mechanisms. Immunobiology. 2007; 212: 475–490. PMID: <u>17544832</u>
- Figueiredo CA, Barreto ML, Rodrigues LC, Cooper PJ, Silva NB, Amorim LD, et al. Chronic Intestinal Helminth Infections Are Associated with Immune Hyporesponsiveness and Induction of a Regulatory Network. Infect Immun. 2010; 78: 3160–3167. doi: 10.1128/IAI.01228-09 PMID: 20404082
- Levine MM. Immunogenicity and efficacy of oral vaccines in developing countries: lessons from a live cholera vaccine. BMC Biol. 2010; 8: 129. doi: <u>10.1186/1741-7007-8-129</u> PMID: <u>20920375</u>
- Serazin AC, Shackelton LA, Wilson C, Bhan MK. Improving the performance of enteric vaccines in the developing world. Nat Immunol. 2010; 11: 769–773. doi: 10.1038/ni0910-769
- Van Essen GA, Beran J, Devaster J-M, Durand C, Duval X, Esen M, et al. Influenza symptoms and their impact on elderly adults: randomised trial of AS03-adjuvanted or non-adjuvanted inactivated trivalent seasonal influenza vaccines. Influenza Other Respir Viruses. 2014; 8: 452–462. doi: <u>10.1111/irv.</u> <u>12245</u> PMID: <u>24702710</u>
- Kaufmann SHE. The contribution of immunology to the rational design of novel antibacterial vaccines. Nat Rev Microbiol. 2007; 5: 491–504. PMID: 17558425
- 20. Almond JW. Vaccine renaissance. Nat Rev Microbiol. 2007; 5: 478–481. PMID: 17571456
- Cooper PJ, Chico ME, Losonsky G, Sandoval C, Espinel I, Sridhara R, et al. Albendazole treatment of children with ascariasis enhances the vibriocidal antibody response to the live attenuated oral cholera vaccine CVD 103-HgR. J Infect Dis. 2000; 182: 1199–1206. PMID: 10979918
- Elias D, Wolday D, Akuffo H, Petros B, Bronner U, Britton S. Effect of deworming on human T cell responses to mycobacterial antigens in helminth-exposed individuals before and after bacille Calmette-Guérin (BCG) vaccination. Clin Exp Immunol. 2001; 123: 219–225. PMID: <u>11207651</u>
- Esen M, Mordmüller B, de Salazar P, Adegnika A, Agnandji S, Schaumburg F, et al. Reduced antibody responses agianst Plasmodium falciparum vaccine candidate antigens in the presence of Trichuris trichiura. Vaccine. 2012; 7621–7624.
- Van den Biggelaar AHJ, Rodrigues LC, van Ree R, van der Zee JS, Hoeksma-Kruize YCM, Souverijn JHM, et al. Long-term treatment of intestinal helminths increases mite skin-test reactivity in Gabonese schoolchildren. J Infect Dis. 2004; 189: 892–900. PMID: 14976607
- Van den Biggelaar AH, Lopuhaa C, van Ree R, van der Zee JS, Jans J, Hoek A, et al. The prevalence of parasite infestation and house dust mite sensitization in Gabonese schoolchildren. Int Arch Allergy Immunol. 2001; 126: 231–238. PMID: 11752881
- Van Riet E, Adegnika AA, Retra K, Vieira R, Tielens AGM, Lell B, et al. Cellular and humoral responses to influenza in gabonese children living in rural and semi-urban areas. J Infect Dis. 2007; 196: 1671– 1678. PMID: <u>18008252</u>
- WHO | Soil-transmitted helminth infections [Internet]. [cited 17 Mar 2014]. <u>http://www.who.int/</u> mediacentre/factsheets/fs366/en/
- Schwarz NG, Grobusch MP, Decker M-L, Goesch J, Poetschke M, Oyakhirome S, et al. WHO 2006 child growth standards: implications for the prevalence of stunting and underweight-for-age in a birth cohort of Gabonese children in comparison to the Centers for Disease Control and Prevention 2000 growth charts and the National Center for Health Statistics 1978 growth references. Public Health Nutr. 2008; 11.
- 29. WHO. STH- Soil- transmitted helminthiases- Eliminating soil-transmitted helminthiases as a public health problem in children- progress report 2001–2010 and strategic plan 2011–2020. WHO Library Cataloguing in Publication Data; 2012.
- Adegnika AA, Zinsou JF, Issifou S, Ateba-Ngoa U, Kassa RF, Feugap EN, et al. Randomized, Controlled, Assessor-Blind Clinical Trial To Assess the Efficacy of Single- versus Repeated-Dose Albendazole To Treat Ascaris lumbricoides, Trichuris trichiura, and Hookworm Infection. Antimicrob Agents Chemother. 2014; 58: 2535–2540. doi: 10.1128/AAC.01317-13 PMID: 24550339
- Rowe T, Abernathy RA, Hu-Primmer J, Thompson WW, Lu X, Lim W, et al. Detection of antibody to avian influenza A (H5N1) virus in human serum by using a combination of serologic assays. J Clin Microbiol. 1999; 37: 937–943. PMID: <u>10074505</u>
- Esen M, Kremsner PG, Schleucher R, Gässler M, Imoukhuede EB, Imbault N, et al. Safety and immunogenicity of GMZ2—a MSP3-GLURP fusion protein malaria vaccine candidate. Vaccine. 2009; 27: 6862–6868. doi: <u>10.1016/j.vaccine.2009.09.011</u> PMID: <u>19755144</u>
- Hauser AE, Debes GF, Arce S, Cassese G, Hamann A, Radbruch A, et al. Chemotactic responsiveness toward ligands for CXCR3 and CXCR4 is regulated on plasma blasts during the time course of a memory immune response. J Immunol Baltim Md 1950. 2002; 169: 1277–1282. PMID: 12133949
- Bernasconi NL, Traggiai E, Lanzavecchia A. Maintenance of serological memory by polyclonal activation of human memory B cells. Science. 2002; 298: 2199–2202. PMID: <u>12481138</u>

- Sapero JJ, Lawless DK. The MIF stain-preservation technic for the identification of intestinal protozoa. Am J Trop Med Hyg. 1953; 2: 613–619. PMID: <u>13065626</u>
- Feldmeier H, Bienzle U, Dietrich M, Sievertsen HJ. Combination of a viability test and a quantification method for Schistosoma haematobium eggs (filtration—trypan blue-staining-technique). Tropenmed Parasitol. 1979; 30: 417–422. PMID: <u>94193</u>
- **37.** R development core team. R: A language and environment for statistical computing [Internet]. Vienna, Austria; 2010. <u>http://www.R-project.org/</u>
- Elliott AM, Kizza M, Quigley MA, Ndibazza J, Nampijja M, Muhangi L, et al. The impact of helminths on the response to immunization and on the incidence of infection and disease in childhood in Uganda: design of a randomized, double-blind, placebo-controlled, factorial trial of deworming interventions delivered in pregnancy and early childhood [ISRCTN32849447]. Clin Trials. 2007; 4: 42–57. PMID: 17327245
- Webb EL, Mawa PA, Ndibazza J, Kizito D, Namatovu A, Kyosiimire-Lugemwa J, et al. Effect of singledose anthelmintic treatment during pregnancy on an infant's response to immunisation and on susceptibility to infectious diseases in infancy: a randomised, double-blind, placebo-controlled trial. Lancet. 2011; 377: 52–62. doi: 10.1016/S0140-6736(10)61457-2 PMID: 21176950
- 40. Pulendran B, Ahmed R. Immunological mechanisms of vaccination. Nat Immunol. 2011; 131: 509–517.
- Nauta JJP, Beyer WEP, Osterhaus ADME. On the relationship between mean antibody level, seroprotection and clinical protection from influenza. Biol J Int Assoc Biol Stand. 2009; 37: 216–221.
- Beyer WEP, Palache AM, Lüchters G, Nauta J, Osterhaus ADME. Seroprotection rate, mean fold increase, seroconversion rate: which parameter adequately expresses seroresponse to influenza vaccination? Virus Res. 2004; 103: 125–132. PMID: 15163500
- 43. Muramatsu M, Yoshida R, Yokoyama A, Miyamoto H, Kajihara M, Maruyama J, et al. Comparison of antiviral activity between IgA and IgG specific to influenza virus hemagglutinin: increased potential of IgA for heterosubtypic immunity. PloS One. 2014; 9: e85582. doi: <u>10.1371/journal.pone.0085582</u> PMID: <u>24465606</u>
- 44. Renegar KB, Small PA. Immunoglobulin A mediation of murine nasal anti-influenza virus immunity. J Virol. 1991; 65: 2146–2148. PMID: 2002558
- Lamm ME. Interaction of antigens and antibodies at mucosal surfaces. Annu Rev Microbiol. 1997; 51: 311–340. PMID: <u>9343353</u>
- Tamura S, Hasegawa H, Kurata T. Estimation of the effective doses of nasal-inactivated influenza vaccine in humans from mouse-model experiments. Jpn J Infect Dis. 2010; 63: 8–15. PMID: 20093755
- Cooper PJ, Chico M, Sandoval C, Espinel I, Guevara A, Levine MM, et al. Human infection with Ascaris lumbricoides is associated with suppression of the interleukin-2 response to recombinant cholera toxin B subunit following vaccination with the live oral cholera vaccine CVD 103-HgR. Infect Immun. 2001; 69: 1574–1580. PMID: <u>11179329</u>
- 48. Nikolay B, Brooker SJ, Pullan RL. Sensitivity of diagnostic tests for human soil-transmitted helminth infections: a meta-analysis in the absence of a true gold standard. Int J Parasitol. 2014.
- **49.** Bawden MP. Improvement of the merthiolate-iodine-formalin (MIF) fecal technique for hookworm, Trichuris trichiura, and Ascaris lumbricoides eggs. J Parasitol. 1994; 80: 474–475. PMID: <u>8195951</u>
- 50. Macpherson CN. The chancing picture of intestinal parasites in the caribbean. caribbean health. 2001; 1: 10–12.
- 51. Adedayo O, Nasiiro R. Intestinal parasitoses. J Natl Med Assoc. 2004; 96: 93–96. PMID: 14746358
- Jia T-W, Melville S, Utzinger J, King CH, Zhou X-N. Soil-transmitted helminth reinfection after drug treatment: a systematic review and meta-analysis. PLoS Negl Trop Dis. 2012; 6: e1621. doi: <u>10.1371/</u> journal.pntd.0001621 PMID: <u>22590656</u>